Author's response to reviews

Title: The Disabilities of the Arm, Shoulder and Hand Questionnaire (DASH) can measure the impairment, activity limitations and participation restriction constructs from the International Classification of Functioning, Disability and Health (ICF)

Authors:

Diane Dixon (diane.dixon@stir.ac.uk)
Marie Johnston (m.johnston@abdn.ac.uk)
Margaret McQueen (mmcqueen@staffmail.ed.ac.uk)
Charles Court-Brown (cortbrown@aol.com)

Version: 3 Date: 8 August 2008

Author's response to reviews: see over
Dear Editor
RE: MS : 3257260061917667
Authors: Diane Dixon, Marie Johnston, Margaret McQueen and Charles Court-Brown
Title: The Disabilities of the Arm, Shoulder and Hand Questionnaire (DASH) can measure the impairment, activity limitations and participation restriction constructs from the International Classification of Functioning, Disability and Health (ICF)
Journal: BMC Musculoskeletal Disorders

The following changes have been made to the above article:
1. The Table placement text has been removed
2. Authors Contribution section now states that all authors read and approved the final manuscript.
3. Two horizontal lines have been added to Table 1 to separate the definitions of each ICF construct provided
4. The asterisk denoting the p-value for a significant match in the footnote to Table 2 is no longer superscripted. This makes this asterisk consistent with others used in Table 2.
5. A line has been inserted between paragraph 1 and 2 of the discussion

Yours faithfully

Diane Dixon